MEG Management Committee meeting

29.06.2009, 14:00hrs, West Wing Seminar Room, CBU

Minutes
· Present:  Friedemann Pulvermuller, Yury Shtyrov, Gary Chandler, Oleg Korzyukov, Olaf Hauk, Rhodri Cusack, Marie Smith, Rob Anthony, Clare Cook, Georgie Morrill.
1. Apologies for absences
· Rik Henson
· Maarten Van Casteren

· Danny Mitchell

2. Matters arising 
FP requested that GM inform reception about the MEG MC meetings to get them listed in the weekly diary.

3. Minutes of the last meeting
Approved with a few minor changes.
Memory in the user room: RA reported that when you fire up a VNC session on your PC, it is running from one of the computers in the server room.  It was a heavy period of use, but Computing are still purchasing more memory to stop this from happening.
The issue with the Trigger values is ongoing.

4. Computing
The memory which has been purchased should be with us this week.
The megadmin account has been created and RA just needs to liaise with Claire about permissions.
RH asked when the new blade systems might be online.  RA informed the committee that this should be sorted within the next month or so.  

RH also asked about the new versions of Matlab and whether they were ready to roll out onto the machines.  RA will be installing version 2009A when building the new machines and it was agreed that only the one version would be needed which would help with disc space.  
RH reported that new version of Maxfilter is still under FDA approval so won’t be out just yet.  FP enquired whether we can run the old version and new side-by-side in case there are any bugs with the new version. This can be tested when it comes out.
FP brought up the fact that 2 researchers who are no longer here are having problems accessing the system to try and analyse their data despite having accounts.  RA wasn’t aware of any problems but will look into.  YS suggested that it could be related to the upgrade of NME scripts.  OK also suggested that perhaps they need to have the GLX session which RA will look into.
Action:  RA to check NME versions and liaise with FP/Veronique Boulenger about remote access issues.
5. Lab business and progress
5.1. Phantom/empty room recording data: OH found that in the 2 recordings they did with supine position, there was no 25Hz interference – this only happened in the upright position.  When they turned off the air con, it ceased.  GC reported that the global air con fan ran at a constant speed.  OH will test again after the next helium refill.
5.2. MEG Admin network space: Discussed in Computing item.  RA to sort permissions for committee members.
5.3. Other issues: YS updated the committee on the fact the helium pipe was broken Helsinki assured YS that it was not dangerous.  There is a risk that this can happen regularly.
OK informed the committee that we have new gold-plated electrodes which have been tested and are all fine.  OK also suggested that the coils should be checked every Wednesday.  YS suggested contacting David in Helsinki as they have an easy way of checking coils.
YS reported that Elekta are happy with the wall construction around the Christmas shut-down period.  Will wait for further updates.
Action: CD to order more grass gel, OK to speak to David in Helsinki
6. MEG publications output
GM showed the draft list of all published papers she has found so far which used the MEG lab.  This is ongoing and will be brought up at the next meeting.
Action: GM to liaise with researchers about publications.
7. MEG Lab stats for June
GM showed the committee the spreadsheet which showed lower usage of the lab, mainly due to GSK cancelling most of their slots and a few others cancelled by another researcher due to not being able to manage to do two slots in one day.  Things look better for July.
8. AOB
FP brought up the fact that it would be very good to have database where you could access all data acquired with all subjects, for example, so that you can see if your subject has had a structural etc.  There would be issues with data protection and computing facilities for this.  This is ongoing but will be difficult to arrange.

9. Date of next meeting
Wednesday 5th August in the WWSR

